Author’s response to reviews

Title: Development of a New Valid and Reliable Microsurgical Skill Assessment Scale for Ophthalmology Residents

Authors:

Zhihua Zhang (rabbit.maomao@gmail.com)
Minwen Zhou (zmw8008@163.com)
Kun Liu (driukun@sjtu.edu.cn)
Bijun Zhu (judyzhu52@163.com)
Haiyun Liu (rabbit_maomao@126.com)
Xiaodong Sun (xdsun@sjtu.edu.cn)
Xun Xu (drxuxun@sjtu.edu.cn)

Version: 2 Date: 24 Feb 2018

Author’s response to reviews:

Dear Prof. Tu:

Thank you for your comments concerning our manuscript entitled “Development of a New Valid and Reliable Microsurgical Skill Assessment Scale for Ophthalmology Residents” (Manuscript No. BOPH-D-17-00405R1). Those comments are all valuable and very helpful for our paper. We have revised the manuscript according to those comments.

Technical Comments:

1. Please provide us the clean final version without track changes.

Response: We have provided the clean final version without track changes.
2. Please kindly provide all authors' emails on title page.
Response: We have provided all author's emails on the title page.

3. Abstract: the subtitle "Purpose" should be Background.
Response: We have changed the subtitle "Purpose" to "Background"

4. Please add a subtitle "Background" for background section.
Response: We have added a subtitle "Background" for the background section.